\begin{landscape}
%%\begin{table*}
\begin{center}
\begin{scriptsize}
\tablefirsthead{\hline
\hline
PSR	&	PSRJ	& $\frac{\sqrt{\dot{E}}}{d^2}$ & $\dot{E}$ 	& $d$  & $S_{1400}$	& $S_{400}$ & Cluster, & Optical & EGRET & EgretDist ($^\circ$) &  TeV & Notes 		\\
&		& \% of Vela 	&(erg/s)	& (kpc)	& (mJy)	& (mJy) & Galaxy & X-ray & nearby & & assoc. &	\\
\hline}
\tablehead{\multicolumn{13}{l}{\small \sl continued from previous page}\\
\hline
\hline
PSR	&	PSRJ	& $\frac{\sqrt{\dot{E}}}{d^2}$ & $\dot{E}$ 	& $d$  & $S_{1400}$	& $S_{400}$ & Cluster, & Optical & EGRET & EgretDist ($^\circ$) &  TeV & Notes 		\\
&		& \% of Vela 	&(erg/s)	& (kpc)	& (mJy)	& (mJy) & Galaxy & X-ray & nearby & & assoc. &	\\
\hline}

\bottomcaption{
Pulsars being timed for the GLAST Large Area Telescope (all known pulsars with $\dot E > 10^{34}$ ergs/s), ordered by
$\sqrt{\dot E}/d^2$, where $\dot E$ is the spin-down energy loss rate and $d$ is the distance. $\sqrt{\dot E}/d^2$ as an indicator of
expected gamma flux suffers many large uncertainties (see text). $S_{1400}$ and $S_{400}$ are the radio flux densities at $1400$ MHz and
$400$ MHz, respectively.``Cluster, Galaxy'' is the name of the globular cluster or the host galaxy, if the pulsar is in one. 
``np'' means that the source is observed in X-rays, but not pulsed. If the pulsar is located less than $2^\circ$ 
away from a $3^{rd}$ EGRET catalog source \citep{3rdcatalog}, the EGRET name and the angular distance are listed. 
The asterisk (*) indicates that the 3EG source has more than one possible counterpart in the table. 
The TeV associations are taken from \citet{HESScatalog}, available at
http://www.mpi-hd.mpg.de/hfm/HESS/public/HESS\_catalog.htm, and from MILAGRO \citep{Milagro}. 
Nearby pulsar wind nebulae (PWN) are noted in the last column \citep{KRH04,MSER04}. 
The data in the first 8 columns were obtained from the ATNF database except for the radio flux
densities with the superscripts \textit{a}: \citet{JBK03}, \textit{b}: \citet{KCM98}, \textit{c}: \citet{RTJ96}, \textit{d}:
\citet{DALM01}, \textit{e}: \citet{CLF00}.  An electronic version of this table is posted at
https://confluence.slac.stanford.edu/display/GLAMCOG/Pulsars+being+timed .}
\tabletail{\multicolumn{13}{r}{\small \sl continuing next page\ldots}\\}
%
\tablelasttail{\hline}
\par
\begin{supertabular}{ccccccccccccc}
\hline
\hline
B0833$-$45 & J0835$-$4510 & 100. & 6.9e+36 & 0.3 & 1100. & 5000. &    & OX & 3EG J0834$-$4511 & 0.34 & Vela & G263.9$-$3.3, Vela X \\ 
J0633+1746 & J0633+1746 & 22. & 3.3e+34 & 0.2 & --- & --- &    & OX & 3EG J0633+1751 & 0.24 &   & G195.1+4.3, Geminga \\ 
B0531+21 & J0534+2200 & 17. & 4.6e+38 & 2.0 & 14. & 646. &    & OX & 3EG J0534+2200 & 0.13 & Crab & G184.6$-$5.8, Crab, SN1054 \\ 
J0437$-$4715 & J0437$-$4715 & 14. & 1.2e+34 & 0.2 & 142. & 550. &    &  X &   &   &   & G253.4$-$42.0 \\ 
B0656+14 & J0659+1414 & 7.42 & 3.8e+34 & 0.3 & 3.70 & 6.50 &    & OX &   &   &   & Monogem Ring \\ 
B0743$-$53 & J0745$-$5353 & 5.37 & 1.1e+34 & 0.2 & --- & 23. &    &    &   &   &   &   \\ 
J0034$-$0534 & J0034$-$0534 & 1.90 & 3.0e+34 & 0.5 & 0.61 & 17. &    &    &   &   &   &   \\ 
J0205+6449 & J0205+6449 & 1.62 & 2.7e+37 & 3.2 & 0.04 & --- &    &  X &   &   &   & G130.7+3.1, 3C 58, SN1181 \\ 
J0613$-$0200 & J0613$-$0200 & 1.58 & 1.3e+34 & 0.5 & 1.40 & 21. &    &    &   &   &   &   \\ 
J1747$-$2958 & J1747$-$2958 & 1.25 & 2.5e+36 & 2.0 & 0.25 & --- &    &    & 3EG J1744$-$3011 (*) & 0.84 & HESS J1745$-$303 & G359.23$-$0.82, Mouse \\ 
B1706$-$44 & J1709$-$4429 & 1.11 & 3.4e+36 & 2.3 & 7.30 & 25. &    &  X & 3EG J1710$-$4439 & 0.18 &   & G343.1$-$2.3 \\ 
B1055$-$52 & J1057$-$5226 & 1.06 & 3.0e+34 & 0.7 & --- & 80. &    &  X & 3EG J1058$-$5234 & 0.12 &   &   \\ 
J1833$-$1034 & J1833$-$1034 & 1.00 & 3.4e+37 & 4.3 & 0.07 & --- &    &    &   &   & HESS J1833$-$105 & G21.5$-$0.9 \\ 
J1740+1000 & J1740+1000 & 0.99 & 2.3e+35 & 1.2 & 9.20 & 3.10 &    &    &   &   &   &   \\ 
B1951+32 & J1952+3252 & 0.98 & 3.7e+36 & 2.5 & 1.00 & 7.00 &    &    &   &   &   & G69.0+2.7, CTB 80 \\ 
J1357$-$6429 & J1357$-$6429 & 0.90 & 3.1e+36 & 2.5 & 0.44 & --- &    &  X &   &   &   &   \\ 
B1509$-$58 & J1513$-$5908 & 0.76 & 1.8e+37 & 4.2 & 0.94 & 1.50 &    &  X &   &   & HESS J1514$-$591 & G320.4$-$1.2, MSH 15-52 \\ 
B1257+12 & J1300+1240 & 0.74 & 1.9e+34 & 0.8 & 2.00 & 20. &    &  np &   &   &   &   \\ 
J1524$-$5625 & J1524$-$5625 & 0.73 & 3.2e+36 & 2.8 & 0.83 & --- &    &    &   &   &   &   \\ 
J1531$-$5610 & J1531$-$5610 & 0.69 & 9.1e+35 & 2.1 & 0.60 & --- &    &    &   &   &   &   \\ 
B1046$-$58 & J1048$-$5832 & 0.60 & 2.0e+36 & 2.7 & 6.50 & --- &    &    & 3EG J1048$-$5840 & 0.14 &   & G287.4+0.58 \\ 
B0355+54 & J0358+5413 & 0.56 & 4.5e+34 & 1.1 & 23. & 46. &    &    &   &   &   &   \\ 
J0940$-$5428 & J0940$-$5428 & 0.50 & 1.9e+36 & 3.0 & 0.66 & --- &    &    &   &   &   &   \\ 
J1930+1852 & J1930+1852 & 0.44 & 1.2e+37 & 5.0 & 0.06 & --- &    &  X & 3EG J1928+1733 & 1.46 &   & G54.1+0.3 \\ 
B1259$-$63 & J1302$-$6350 & 0.37 & 8.2e+35 & 2.8 & 1.70 & --- &    &  np &   &   & HESS J1302$-$638 &   \\ 
J0834$-$4159 & J0834$-$4159 & 0.36 & 9.9e+34 & 1.7 & 0.19 & --- &    &    &   &   &   &   \\ 
J1909$-$3744 & J1909$-$3744 & 0.36 & 2.2e+34 & 1.1 & --- & { 3.}$^{a}$ &    &    &   &   &   &   \\ 
B0906$-$49 & J0908$-$4913 & 0.35 & 4.9e+35 & 2.5 & 10. & 28. &    &    &   &   &   & G270.3$-$1.0 \\ 
J1509$-$5850 & J1509$-$5850 & 0.34 & 5.2e+35 & 2.6 & 0.15 & --- &    &    &   &   &   &   \\ 
B1823$-$13 & J1826$-$1334 & 0.34 & 2.8e+36 & 3.9 & 2.10 & --- &    &  np & 3EG J1826$-$1302 (*) & 0.55 & HESS J1825$-$137 & G18.0$-$0.7 \\ 
J1809$-$1917 & J1809$-$1917 & 0.34 & 1.8e+36 & 3.5 & 2.50 & --- &    &  np &   &   & HESS J1809$-$193 &   \\ 
J0538+2817 & J0538+2817 & 0.32 & 4.9e+34 & 1.5 & 1.90 & 8.20 &    &  X &   &   &   & S147 \\ 
J1811$-$1925 & J1811$-$1925 & 0.32 & 6.4e+36 & 5.0 & --- & --- &    &  X &   &   & HESS J1809$-$193 & G11.2$-$0.3, SN 386 \\ 
J1420$-$6048 & J1420$-$6048 & 0.31 & 1.0e+37 & 5.6 & 0.90 & --- &    &  X & 3EG J1420$-$6038 & 0.17 & HESS J1420$-$607 & G313.6+0.3, Kookaburra \\ 
B1800$-$21 & J1803$-$2137 & 0.31 & 2.2e+36 & 3.9 & 7.60 & 23. &    &  np &   &   & HESS J1804$-$216 &   \\ 
J1046+0304 & J1046+0304 & 0.30 & 1.4e+34 & 1.1 & 0.30 & --- &    &    &   &   &   &   \\ 
B0114+58 & J0117+5914 & 0.30 & 2.2e+35 & 2.2 & 0.30 & 7.60 &    &    &   &   &   &   \\ 
J2229+6114 & J2229+6114 & 0.28 & 2.2e+37 & 7.2 & 0.25 & --- &    &  X & 3EG J2227+6122 & 0.54 &   & G106.6+3.1 \\ 
J1718$-$3825 & J1718$-$3825 & 0.28 & 1.2e+36 & 3.6 & 1.30 & --- &    &    & 3EG J1714$-$3857 & 1.18 & HESS J1718$-$385 &   \\ 
B1727$-$33 & J1730$-$3350 & 0.27 & 1.2e+36 & 3.5 & 3.20 & 9.20 &    &    & 3EG J1734$-$3232 & 1.57 &   &   \\ 
B0740$-$28 & J0742$-$2822 & 0.27 & 1.4e+35 & 2.1 & 15. & 296. &    &    &   &   &   &   \\ 
J1617$-$5055 & J1617$-$5055 & 0.27 & 1.6e+37 & 6.8 & { 0.5}$^{b}$ & --- &    &  X &   &   & HESS J1616$-$508 &   \\ 
J1843$-$1113 & J1843$-$1113 & 0.27 & 6.0e+34 & 1.7 & 0.10 & --- &    &    &   &   &   &   \\ 
J2129$-$5721 & J2129$-$5721 & 0.26 & 2.3e+34 & 1.4 & 1.40 & 14. &    &    &   &   &   &   \\ 
J1124$-$5916 & J1124$-$5916 & 0.26 & 1.2e+37 & 6.5 & 0.08 & --- &    &  X &   &   &   & G292.0+1.8, MSH 11$-$54 \\ 
J1846$-$0258 & J1846$-$0258 & 0.25 & 8.1e+36 & 6.0 & --- & --- &    &  X &   &   & HESS J1846$-$029 & G29.7$-$0.3, Kes 75 \\ 
J1913+1011 & J1913+1011 & 0.24 & 2.9e+36 & 4.8 & 0.50 & --- &    &    &   &   & HESS J1912+101 &   \\ 
J1911$-$1114 & J1911$-$1114 & 0.23 & 1.2e+34 & 1.2 & 0.50 & 31. &    &    & 3EG J1904$-$1124 & 1.96 &   &   \\ 
J2043+2740 & J2043+2740 & 0.23 & 5.6e+34 & 1.8 & --- & {15.}$^{c}$ &    &  X &   &   &   &   \\ 
J0855$-$4644 & J0855$-$4644 & 0.22 & 1.1e+36 & 3.9 & 0.20 & --- &    &    &   &   &   &   \\ 
J0218+4232 & J0218+4232 & 0.22 & 2.4e+35 & 2.7 & 0.90 & 35. &    &  X & 3EG J0222+4253 & 1.03 &   &   \\ 
J1739$-$3023 & J1739$-$3023 & 0.20 & 3.0e+35 & 2.9 & 1.00 & --- &    &    & 3EG J1744$-$3011 (*) & 1.10 &   &   \\ 
J1831$-$0952 & J1831$-$0952 & 0.20 & 1.1e+36 & 4.0 & 0.33 & --- &    &    &   &   &   &   \\ 
B1957+20 & J1959+2048 & 0.20 & 1.6e+35 & 2.5 & 0.40 & 20. &    &  np &   &   &   & G59.2$-$4.7 \\ 
J1105$-$6107 & J1105$-$6107 & 0.20 & 2.5e+36 & 5.0 & 0.75 & --- &    &  np & 3EG J1102$-$6103 & 0.86 &   & MSH 11$-$62 \\ 
B1821$-$24 & J1824$-$2452 & 0.19 & 2.2e+36 & 4.9 & 0.18 & 40. & M28  &  X &   &   &   &   \\ 
B1853+01 & J1856+0113 & 0.19 & 4.3e+35 & 3.3 & 0.19 & 3.40 &    &    & 3EG J1856+0114 & 0.05 &   & G34.7$-$0.4, W44, 3C 392 \\ 
B1757$-$24 & J1801$-$2451 & 0.18 & 2.6e+36 & 5.2 & 0.85 & 7.80 &    &  np & 3EG J1800$-$2338 & 1.26 &   & G5.27$-$0.9,G5.4-1.2? \\ 
B0611+22 & J0614+2229 & 0.18 & 6.2e+34 & 2.1 & 2.20 & 29. &    &    & 3EG J0617+2238 & 0.69 &   &   \\ 
B1719$-$37 & J1722$-$3712 & 0.16 & 3.3e+34 & 1.9 & 3.20 & 25. &    &    &   &   &   &   \\ 
J1835$-$1106 & J1835$-$1106 & 0.16 & 1.8e+35 & 2.8 & 2.20 & 30. &    &    &   &   &   &   \\ 
B0540+23 & J0543+2329 & 0.15 & 4.1e+34 & 2.1 & 9.00 & 29. &    &    &   &   &   &   \\ 
J1913+0904 & J1913+0904 & 0.14 & 1.6e+35 & 3.0 & 0.07 & --- &    &    &   &   &   &   \\ 
J0857$-$4424 & J0857$-$4424 & 0.13 & 2.6e+34 & 1.9 & 0.88 & 12. &    &    &   &   &   &   \\ 
J0729$-$1448 & J0729$-$1448 & 0.13 & 2.8e+35 & 3.5 & 0.70 & --- &    &    &   &   &   &   \\ 
B1737$-$30 & J1740$-$3015 & 0.12 & 8.2e+34 & 2.7 & 6.40 & 25. &    &    & 3EG J1744$-$3011 (*) & 0.86 &   &   \\ 
J1928+1746 & J1928+1746 & 0.11 & 1.6e+36 & 5.8 & 0.25 & --- &    &    & 3EG J1928+1733 & 0.29 &   &   \\ 
J1015$-$5719 & J1015$-$5719 & 0.11 & 8.3e+35 & 5.1 & 0.90 & --- &    &    & 3EG J1014$-$5705 & 0.47 &   &   \\ 
B1830$-$08 & J1833$-$0827 & 0.11 & 5.8e+35 & 4.7 & 3.60 & --- &    &    &   &   & HESS J1834$-$087 &   \\ 
J1837$-$0604 & J1837$-$0604 & 0.11 & 2.0e+36 & 6.4 & 0.70 & --- &    &    & 3EG J1837$-$0606 (*) & 0.18 &   &   \\ 
J1740$-$5340A & J1740$-$5340A & 0.10 & 1.4e+35 & 3.4 & { 1.}$^{d}$ & --- & NGC6397  &    &   &   &   &   \\ 
B1449$-$64 & J1453$-$6413 & 0.10 & 1.9e+34 & 2.1 & 14. & 230. &    &    &   &   &   &   \\ 
J1637$-$4642 & J1637$-$4642 & 0.099 & 6.4e+35 & 5.1 & 0.78 & --- &    &    & 3EG J1639$-$4702 & 0.55 &   &   \\ 
J0631+1036 & J0631+1036 & 0.098 & 1.7e+35 & 3.7 & 0.80 & 1.50 &    &  X &   &   &   &   \\ 
J1702$-$4310 & J1702$-$4310 & 0.096 & 6.3e+35 & 5.1 & 0.72 & --- &    &    &   &   &   &   \\ 
J1301$-$6305 & J1301$-$6305 & 0.094 & 1.7e+36 & 6.7 & 0.46 & --- &    &    &   &   & HESS J1301$-$631 &   \\ 
J1828$-$1101 & J1828$-$1101 & 0.092 & 1.6e+36 & 6.6 & 2.90 & --- &    &    &   &   &   &   \\ 
B1620$-$26 & J1623$-$2631 & 0.091 & 2.0e+34 & 2.2 & 1.60 & 15. & M4  &    & 3EG J1626$-$2519 & 1.35 &   &   \\ 
B1634$-$45 & J1637$-$4553 & 0.087 & 7.5e+34 & 3.2 & 1.10 & 15. &    &    & 3EG J1639$-$4702 & 1.17 &   &   \\ 
J1702$-$4128 & J1702$-$4128 & 0.082 & 3.4e+35 & 4.8 & 1.10 & --- &    &    &   &   & HESS J1702$-$420 &   \\ 
J1705$-$3950 & J1705$-$3950 & 0.081 & 7.4e+34 & 3.3 & 1.50 & --- &    &    &   &   &   &   \\ 
J1016$-$5857 & J1016$-$5857 & 0.080 & 2.6e+36 & 8.0 & 0.46 & --- &    &    & 3EG J1013$-$5915 (*) & 0.89 &   & G284.3$-$1.8 \\ 
J0901$-$4624 & J0901$-$4624 & 0.080 & 4.0e+34 & 2.8 & 0.46 & --- &    &    &   &   &   &   \\ 
B2334+61 & J2337+6151 & 0.080 & 6.2e+34 & 3.1 & 1.40 & 10. &    &  np &   &   &   &   \\ 
B1317$-$53 & J1320$-$5359 & 0.076 & 1.7e+34 & 2.3 & --- & 18. &    &    & 3EG J1316$-$5244 & 1.75 &   &   \\ 
J1841$-$0345 & J1841$-$0345 & 0.073 & 2.7e+35 & 4.8 & 1.40 & --- &    &    & 3EG J1837$-$0423 & 1.29 &   &   \\ 
J1830$-$0131 & J1830$-$0131 & 0.069 & 2.3e+34 & 2.6 & 0.35 & --- &    &    &   &   &   &   \\ 
J1119$-$6127 & J1119$-$6127 & 0.068 & 2.3e+36 & 8.4 & 0.80 & --- &    &  np &   &   &   & G292.2$-$0.5 \\ 
J1549$-$4848 & J1549$-$4848 & 0.068 & 2.3e+34 & 2.7 & 0.47 & 17. &    &    &   &   &   &   \\ 
J1816$-$0755 & J1816$-$0755 & 0.065 & 2.5e+34 & 2.8 & 0.17 & --- &    &    &   &   &   &   \\ 
J1857+0143 & J1857+0143 & 0.064 & 4.5e+35 & 5.8 & 0.74 & --- &    &    & 3EG J1856+0114 & 0.63 & HESS J1858+020 &   \\ 
B1001$-$47 & J1003$-$4747 & 0.064 & 3.0e+34 & 2.9 & --- & 6.00 &    &    &   &   &   &   \\ 
B1610$-$50 & J1614$-$5048 & 0.064 & 1.6e+36 & 7.9 & 2.40 & --- &    &    &   &   &   &   \\ 
B1828$-$11 & J1830$-$1059 & 0.060 & 3.6e+34 & 3.2 & 1.40 & 2.10 &    &    &   &   &   &   \\ 
J1648$-$4611 & J1648$-$4611 & 0.059 & 2.1e+35 & 5.0 & 0.58 & --- &    &    & 3EG J1655$-$4554 & 1.68 &   &   \\ 
B1643$-$43 & J1646$-$4346 & 0.057 & 3.6e+35 & 5.8 & 0.98 & --- &    &    &   &   &   & G341.2+0.9 \\ 
J1906+0746 & J1906+0746 & 0.057 & 2.7e+35 & 5.4 & 0.55 & 0.90 &    &    &   &   &   &   \\ 
B0136+57 & J0139+5814 & 0.055 & 2.1e+34 & 2.9 & 4.60 & 28. &    &    &   &   &   &   \\ 
J1738$-$2955 & J1738$-$2955 & 0.050 & 3.7e+34 & 3.5 & 0.29 & --- &    &    & 3EG J1736$-$2908 (*) & 1.07 &   &   \\ 
B1730$-$37 & J1733$-$3716 & 0.050 & 1.5e+34 & 2.8 & 3.40 & --- &    &    &   &   &   &   \\ 
B0021$-$72F & J0024$-$7204F & 0.049 & 1.4e+35 & 4.9 & { 0.15}$^{e}$ & --- & 47Tuc  &    &   &   &   &   \\ 
J1723$-$3659 & J1723$-$3659 & 0.049 & 3.8e+34 & 3.5 & 1.50 & --- &    &    &   &   &   &   \\ 
J1715$-$3903 & J1715$-$3903 & 0.049 & 6.9e+34 & 4.1 & 0.46 & --- &    &    & 3EG J1714$-$3857 & 0.33 &   &   \\ 
J1601$-$5335 & J1601$-$5335 & 0.048 & 1.0e+35 & 4.5 & 0.22 & --- &    &    &   &   &   &   \\ 
J1514$-$5925 & J1514$-$5925 & 0.048 & 3.5e+34 & 3.5 & 0.27 & --- &    &    &   &   & HESS J1514$-$591 &   \\ 
B1937+21 & J1939+2134 & 0.048 & 1.1e+36 & 8.3 & 10. & 240. &    &  X &   &   &   &   \\ 
B1820$-$30A & J1823$-$3021A & 0.046 & 8.3e+35 & 7.9 & 0.72 & 16. & NGC6624  &    &   &   &   &   \\ 
J1828$-$1057 & J1828$-$1057 & 0.046 & 5.5e+34 & 4.0 & 0.23 & --- &    &    &   &   &   &   \\ 
J0905$-$5127 & J0905$-$5127 & 0.045 & 2.4e+34 & 3.3 & 1.10 & 12. &    &    &   &   &   &   \\ 
J1112$-$6103 & J1112$-$6103 & 0.045 & 4.5e+36 & 12.2 & 1.40 & --- &    &    &   &   &   &   \\ 
B1221$-$63 & J1224$-$6407 & 0.044 & 1.9e+34 & 3.1 & 3.90 & 48. &    &    &   &   &   &   \\ 
B1727$-$47 & J1731$-$4744 & 0.042 & 1.1e+34 & 2.8 & 12. & 190. &    &    &   &   &   &   \\ 
B1356$-$60 & J1359$-$6038 & 0.041 & 1.2e+35 & 5.2 & 7.60 & 105. &    &    &   &   &   &   \\ 
J1921+0812 & J1921+0812 & 0.040 & 2.3e+34 & 3.5 & 0.66 & --- &    &    &   &   &   &   \\ 
B1915+13 & J1917+1353 & 0.039 & 3.8e+34 & 4.0 & 1.90 & 43. &    &    &   &   &   &   \\ 
B0021$-$72E & J0024$-$7204E & 0.039 & 8.8e+34 & 4.9 & { 0.21}$^{e}$ & --- & 47Tuc  &    &   &   &   &   \\ 
J1838$-$0549 & J1838$-$0549 & 0.039 & 1.0e+35 & 5.1 & 0.29 & --- &    &    & 3EG J1837$-$0606 (*) & 0.50 & HESS J1841$-$055 &   \\ 
J1548$-$5607 & J1548$-$5607 & 0.039 & 8.5e+34 & 4.9 & 1.00 & --- &    &    &   &   &   &   \\ 
J1835$-$0944 & J1835$-$0944 & 0.039 & 5.6e+34 & 4.4 & 0.41 & --- &    &    &   &   &   &   \\ 
J2021+3651 & J2021+3651 & 0.038 & 3.4e+36 & 12.4 & 0.10 & --- &    &  X & 3EG J2021+3716 (*) & 0.41 & MGRO J2019+37 & G75.2+0.1 \\ 
B1535$-$56 & J1539$-$5626 & 0.037 & 1.3e+34 & 3.1 & 4.60 & --- &    &    &   &   &   &   \\ 
J1638$-$4608 & J1638$-$4608 & 0.036 & 9.4e+34 & 5.2 & 0.33 & --- &    &    & 3EG J1639$-$4702 & 0.91 &   &   \\ 
J1910$-$5959D & J1910$-$5959D & 0.036 & 5.2e+34 & 4.5 & 0.05 & --- & NGC6752  &    &   &   &   &   \\ 
J1841$-$0524 & J1841$-$0524 & 0.035 & 1.0e+35 & 5.3 & 0.20 & --- &    &    & 3EG J1837$-$0606 (*) & 1.39 & HESS J1841$-$055 &   \\ 
J0024$-$7204O & J0024$-$7204O & 0.033 & 6.5e+34 & 4.9 & { 0.10}$^{e}$ & --- & 47Tuc  &    &   &   &   &   \\ 
B1754$-$24 & J1757$-$2421 & 0.033 & 4.0e+34 & 4.4 & 3.90 & 20. &    &    & 3EG J1800$-$2338 & 0.97 &   &   \\ 
J1452$-$5851 & J1452$-$5851 & 0.032 & 3.5e+34 & 4.3 & 0.24 & --- &    &    &   &   &   &   \\ 
J1541$-$5535 & J1541$-$5535 & 0.032 & 1.1e+35 & 5.7 & 0.22 & --- &    &    &   &   &   &   \\ 
J1643$-$4505 & J1643$-$4505 & 0.031 & 9.4e+34 & 5.6 & 0.28 & --- &    &    &   &   &   &   \\ 
B1607$-$52 & J1611$-$5209 & 0.031 & 3.4e+34 & 4.3 & 1.20 & --- &    &    &   &   &   &   \\ 
B1338$-$62 & J1341$-$6220 & 0.030 & 1.4e+36 & 11.1 & 1.90 & --- &    &  X &   &   &   & G308.8$-$0.1 \\ 
J1907+0918 & J1907+0918 & 0.029 & 3.2e+35 & 7.8 & 0.29 & 0.40 &    &    &   &   &   &   \\ 
J1016$-$5819 & J1016$-$5819 & 0.029 & 4.1e+34 & 4.7 & 0.31 & --- &    &    & 3EG J1013$-$5915 (*) & 1.23 &   &   \\ 
J0537$-$6910 & J0537$-$6910 & 0.029 & 4.9e+38 & 49.4 & --- & --- &  LMC &  X & 3EG J0533$-$6916 & 1.20 &   & G279.6$-$31.7, N 157B \\ 
B1930+22 & J1932+2220 & 0.028 & 7.5e+35 & 9.8 & 1.20 & 7.80 &    &    &   &   &   &   \\ 
J1853$-$0004 & J1853$-$0004 & 0.028 & 2.1e+35 & 7.2 & 0.87 & --- &    &    & 3EG J1856+0114 & 1.46 &   &   \\ 
J0024$-$7204U & J0024$-$7204U & 0.028 & 4.6e+34 & 4.9 & { 0.06}$^{e}$ & --- & 47Tuc  &    &   &   &   &   \\ 
J1019$-$5749 & J1019$-$5749 & 0.028 & 1.8e+35 & 6.9 & 0.80 & --- &    &    & 3EG J1014$-$5705 (*) & 1.64 &   &   \\ 
B1556$-$57 & J1600$-$5751 & 0.027 & 1.1e+34 & 3.5 & 1.40 & 20. &    &    &   &   &   &   \\ 
J1841+0130 & J1841+0130 & 0.027 & 1.2e+34 & 3.6 & 0.06 & --- &    &    &   &   &   &   \\ 
J1543$-$5459 & J1543$-$5459 & 0.026 & 3.8e+34 & 4.8 & 0.62 & --- &    &    &   &   &   &   \\ 
J0954$-$5430 & J0954$-$5430 & 0.026 & 1.6e+34 & 3.9 & 0.36 & --- &    &    &   &   &   &   \\ 
J1737$-$3137 & J1737$-$3137 & 0.026 & 6.0e+34 & 5.5 & 0.80 & --- &    &    & 3EG J1734$-$3232 & 1.19 &   &   \\ 
J1650$-$4921 & J1650$-$4921 & 0.025 & 1.9e+34 & 4.1 & 0.16 & --- &    &    &   &   &   &   \\ 
J1815$-$1738 & J1815$-$1738 & 0.025 & 3.9e+35 & 8.8 & 0.25 & --- &    &    &   &   & HESS J1813$-$178 &   \\ 
B1822$-$14 & J1825$-$1446 & 0.025 & 4.1e+34 & 5.1 & 2.60 & --- &    &    & 3EG J1824$-$1514 (*) & 0.52 & HESS J1826$-$148  &   \\ 
B1821$-$19 & J1824$-$1945 & 0.024 & 3.0e+34 & 4.7 & 4.90 & 71. &    &    &   &   &   &   \\ 
J1853+0056 & J1853+0056 & 0.024 & 4.0e+34 & 5.1 & 0.21 & --- &    &    & 3EG J1856+0114 & 0.68 &   &   \\ 
J1115$-$6052 & J1115$-$6052 & 0.023 & 1.6e+34 & 4.1 & 0.38 & --- &    &    &   &   &   &   \\ 
J1903+0925 & J1903+0925 & 0.023 & 3.2e+34 & 4.9 & 0.20 & --- &    &    &   &   &   &   \\ 
J1413$-$6141 & J1413$-$6141 & 0.023 & 5.6e+35 & 10.1 & 0.61 & --- &    &    & 3EG J1410$-$6147 & 0.80 &   &   \\ 
J1756$-$2225 & J1756$-$2225 & 0.023 & 3.1e+34 & 5.0 & 0.25 & --- &    &    & 3EG J1800$-$2338 & 1.50 &   &   \\ 
J1406$-$6121 & J1406$-$6121 & 0.022 & 2.2e+35 & 8.2 & 0.36 & --- &    &    & 3EG J1410$-$6147 & 0.90 &   &   \\ 
J0024$-$7204T & J0024$-$7204T & 0.021 & 2.7e+34 & 4.9 & --- & --- & 47Tuc  &    &   &   &   &   \\ 
B1718$-$35 & J1721$-$3532 & 0.021 & 4.5e+34 & 5.6 & 11. & --- &    &    &   &   &   &   \\ 
B1508$-$57 & J1512$-$5759 & 0.021 & 1.3e+35 & 7.3 & 6.00 & --- &    &    &   &   &   &   \\ 
B1832$-$06 & J1835$-$0643 & 0.019 & 5.6e+34 & 6.2 & 1.30 & --- &    &    & 3EG J1837$-$0606 & 0.78 &   &   \\ 
B1838$-$04 & J1841$-$0425 & 0.019 & 3.9e+34 & 5.7 & 2.60 & 2.60 &    &    & 3EG J1837$-$0423 (*) & 1.02 &   &   \\ 
J1538$-$5551 & J1538$-$5551 & 0.019 & 1.1e+35 & 7.5 & 0.25 & --- &    &    &   &   &   &   \\ 
J0024$-$7204Q & J0024$-$7204Q & 0.018 & 2.0e+34 & 4.9 & { 0.05}$^{e}$ & --- & 47Tuc  &    &   &   &   &   \\ 
J1138$-$6207 & J1138$-$6207 & 0.018 & 3.0e+35 & 9.7 & 0.49 & --- &    &    &   &   &   &   \\ 
J1156$-$5707 & J1156$-$5707 & 0.018 & 4.3e+34 & 6.0 & 0.19 & --- &    &    &   &   &   &   \\ 
J1734$-$3333 & J1734$-$3333 & 0.018 & 5.6e+34 & 6.5 & 0.50 & --- &    &    & 3EG J1734$-$3232 & 1.03 &   &   \\ 
J1412$-$6145 & J1412$-$6145 & 0.018 & 1.2e+35 & 7.8 & 0.47 & --- &    &    & 3EG J1410$-$6147 & 0.53 &   &   \\ 
J1743$-$3153 & J1743$-$3153 & 0.017 & 5.8e+34 & 6.6 & 0.50 & --- &    &    & 3EG J1744$-$3011 & 1.71 &   &   \\ 
J1632$-$4757 & J1632$-$4757 & 0.017 & 5.0e+34 & 6.4 & 0.30 & --- &    &    & 3EG J1639$-$4702 & 1.92 & HESS J1632$-$478 &   \\ 
J1123$-$6259 & J1123$-$6259 & 0.017 & 1.0e+34 & 4.3 & 0.56 & 11. &    &    &   &   &   &   \\ 
J1909+0912 & J1909+0912 & 0.017 & 1.3e+35 & 8.2 & 0.35 & --- &    &    &   &   &   &   \\ 
J1650$-$4502 & J1650$-$4502 & 0.017 & 1.1e+34 & 4.4 & 0.35 & --- &    &    & 3EG J1655$-$4554 & 1.41 &   &   \\ 
J1638$-$4417 & J1638$-$4417 & 0.016 & 3.9e+34 & 6.2 & 0.21 & --- &    &    &   &   &   &   \\ 
B0540$-$69 & J0540$-$6919 & 0.016 & 1.5e+38 & 49.4 & 0.02 & --- &  LMC & OX & 3EG J0533$-$6916 & 1.80 &   & G279.7$-$31.5, N158A \\ 
J1834$-$0731 & J1834$-$0731 & 0.015 & 1.7e+34 & 5.1 & 1.00 & --- &    &    & 3EG J1837$-$0606 & 1.57 &   &   \\ 
J1052$-$5954 & J1052$-$5954 & 0.015 & 1.3e+35 & 8.5 & 0.15 & --- &    &    & 3EG J1048$-$5840 & 1.70 &   &   \\ 
B2148+52 & J2150+5247 & 0.015 & 1.1e+34 & 4.6 & 2.00 & 16. &    &    &   &   &   &   \\ 
J1649$-$4653 & J1649$-$4653 & 0.015 & 1.1e+34 & 4.7 & 0.31 & --- &    &    & 3EG J1655$-$4554 & 1.71 &   &   \\ 
J1551$-$5310 & J1551$-$5310 & 0.015 & 8.3e+34 & 7.8 & 0.54 & --- &    &    &   &   &   &   \\ 
J1843$-$0702 & J1843$-$0702 & 0.015 & 1.2e+34 & 4.8 & 0.17 & --- &    &    & 3EG J1837$-$0606 & 1.85 &   &   \\ 
J1907+0919 & J1907+0919 & 0.014 & 2.2e+34 & 5.8 & --- & --- &    &    &   &   &   &   \\ 
J1349$-$6130 & J1349$-$6130 & 0.014 & 1.2e+34 & 5.0 & 0.58 & --- &    &    &   &   &   &   \\ 
J1913+0832 & J1913+0832 & 0.014 & 7.4e+34 & 7.9 & 0.60 & --- &    &    &   &   &   &   \\ 
J1838$-$0453 & J1838$-$0453 & 0.013 & 8.3e+34 & 8.1 & 0.33 & --- &    &    & 3EG J1837$-$0423 (*) & 0.59 &   &   \\ 
J1627$-$4706 & J1627$-$4706 & 0.013 & 2.5e+34 & 6.1 & 0.10 & --- &    &    &   &   &   &   \\ 
J1837$-$0559 & J1837$-$0559 & 0.013 & 1.6e+34 & 5.4 & 0.50 & --- &    &    & 3EG J1837$-$0606 (*) & 0.15 & HESS J1841$-$055 &   \\ 
J1248$-$6344 & J1248$-$6344 & 0.013 & 8.6e+34 & 8.3 & 0.12 & --- &    &    &   &   &   &   \\ 
J1845$-$0743 & J1845$-$0743 & 0.013 & 1.3e+34 & 5.2 & 2.70 & --- &    &    &   &   &   &   \\ 
J1839$-$0905 & J1839$-$0905 & 0.012 & 1.4e+34 & 5.4 & 0.16 & --- &    &    &   &   &   &   \\ 
J1845$-$0316 & J1845$-$0316 & 0.012 & 3.9e+34 & 7.0 & 0.35 & --- &    &    &   &   & HESS J1846$-$029 &   \\ 
J1801$-$2154 & J1801$-$2154 & 0.012 & 1.2e+34 & 5.3 & 0.18 & --- &    &    & 3EG J1800$-$2338 & 1.75 &   &   \\ 
J1853+0545 & J1853+0545 & 0.012 & 1.2e+34 & 5.3 & 1.60 & --- &    &    &   &   &   &   \\ 
J1907+0731 & J1907+0731 & 0.011 & 1.5e+34 & 5.7 & 0.35 & --- &    &    &   &   &   &   \\ 
B1557$-$50 & J1600$-$5044 & 0.011 & 2.8e+34 & 6.7 & 17. & --- &    &    &   &   &   &   \\ 
J1755$-$2534 & J1755$-$2534 & 0.011 & 3.5e+34 & 7.1 & 0.17 & --- &    &    &   &   &   &   \\ 
J1452$-$6036 & J1452$-$6036 & 0.011 & 1.5e+34 & 5.8 & 1.40 & --- &    &    &   &   &   &   \\ 
J1839$-$0321 & J1839$-$0321 & 0.011 & 3.6e+34 & 7.2 & 0.27 & --- &    &    & 3EG J1837$-$0423 & 1.22 &   &   \\ 
J1632$-$4818 & J1632$-$4818 & 0.011 & 4.8e+34 & 7.8 & 0.39 & --- &    &    &   &   & HESS J1632$-$478 &   \\ 
B1841$-$05 & J1844$-$0538 & 0.011 & 2.3e+34 & 6.5 & 2.20 & --- &    &    & 3EG J1837$-$0606 & 1.83 &   &   \\ 
J1550$-$5418 & J1550$-$5418 & 0.011 & 1.0e+35 & 9.6 & 3.30 & --- &    &    &   &   &   &   \\ 
B2000+32 & J2002+3217 & 0.010 & 1.2e+34 & 5.7 & 1.20 & 5.50 &    &    &   &   &   &   \\ 
J1907+0345 & J1907+0345 & 0.0094 & 2.3e+34 & 7.2 & 0.17 & --- &    &    &   &   &   &   \\ 
J1842$-$0905 & J1842$-$0905 & 0.0091 & 1.0e+34 & 5.9 & 0.81 & --- &    &    &   &   &   &   \\ 
J1946+2611 & J1946+2611 & 0.0089 & 1.1e+34 & 6.1 & --- & 1.50 &    &    &   &   &   &   \\ 
J1726$-$3530 & J1726$-$3530 & 0.0084 & 3.5e+34 & 8.4 & 0.30 & --- &    &    &   &   &   &   \\ 
B1636$-$47 & J1640$-$4715 & 0.0083 & 1.2e+34 & 6.5 & 1.20 & --- &    &    & 3EG J1639$-$4702 & 0.38 &   &   \\ 
J1904+0800 & J1904+0800 & 0.0082 & 3.7e+34 & 8.6 & 0.36 & --- &    &    &   &   &   &   \\ 
J1820$-$1529 & J1820$-$1529 & 0.0080 & 4.0e+34 & 8.9 & 0.61 & --- &    &    & 3EG J1824$-$1514 & 0.87 &   &   \\ 
B2011+38 & J2013+3845 & 0.0076 & 2.9e+34 & 8.4 & 6.40 & 26. &    &    & 3EG J2016+3657 & 1.95 &   &   \\ 
J1908+0909 & J1908+0909 & 0.0075 & 3.6e+34 & 9.0 & 0.22 & --- &    &    &   &   &   &   \\ 
B1855+02 & J1857+0212 & 0.0074 & 2.2e+34 & 8.0 & 1.60 & --- &    &    & 3EG J1856+0114 & 1.07 & HESS J1858+020 &   \\ 
J1903+0601 & J1903+0601 & 0.0073 & 1.5e+34 & 7.2 & 0.26 & --- &    &    & 3EG J1903+0550 & 0.21 &   &   \\ 
J1515$-$5720 & J1515$-$5720 & 0.0072 & 1.0e+34 & 6.6 & 0.20 & --- &    &    &   &   &   &   \\ 
J1853+0011 & J1853+0011 & 0.0070 & 2.1e+34 & 8.1 & 0.30 & --- &    &    & 3EG J1856+0114 & 1.21 &   &   \\ 
J1342+2822B & J1342+2822B & 0.0068 & 5.4e+34 & 10.4 & 0.01 & --- & M3  &    &   &   &   &   \\ 
J1806$-$2125 & J1806$-$2125 & 0.0068 & 4.3e+34 & 9.8 & 1.10 & --- &    &    &   &   &   &   \\ 
J1626$-$4807 & J1626$-$4807 & 0.0066 & 2.7e+34 & 8.9 & 0.37 & --- &    &    &   &   &   &   \\ 
J1020$-$6026 & J1020$-$6026 & 0.0065 & 9.6e+34 & 12.3 & 0.14 & --- &    &    &   &   &   &   \\ 
J1305$-$6203 & J1305$-$6203 & 0.0055 & 1.6e+34 & 8.5 & 0.62 & --- &    &    & 3EG J1308$-$6112 & 1.08 &   &   \\ 
J1843$-$0355 & J1843$-$0355 & 0.0055 & 1.8e+34 & 8.8 & 0.80 & --- &    &    & 3EG J1837$-$0423 & 1.59 &   &   \\ 
J1735$-$3258 & J1735$-$3258 & 0.0054 & 2.4e+34 & 9.6 & 0.46 & --- &    &    & 3EG J1734$-$3232 & 0.66 &   &   \\ 
B2127+11E & J2129+1210E & 0.0050 & 7.0e+34 & 12.9 & --- & 0.20 & M15  &    &   &   &   &   \\ 
B1758$-$23 & J1801$-$2304 & 0.0050 & 6.2e+34 & 12.6 & 2.20 & --- &    &    & 3EG J1800$-$2338 (*) & 0.65 &   &   \\ 
J1043$-$6116 & J1043$-$6116 & 0.0046 & 1.7e+34 & 9.5 & 0.91 & --- &    &    &   &   &   &   \\ 
J1908+0839 & J1908+0839 & 0.0045 & 1.5e+34 & 9.3 & 0.49 & --- &    &    &   &   &   &   \\ 
J1808$-$2024 & J1808$-$2024 & 0.0041 & 5.0e+34 & 13.1 & --- & --- &    &    &   &   &   &   \\ 
J1812$-$1910 & J1812$-$1910 & 0.0035 & 1.9e+34 & 11.2 & 0.22 & --- &    &    &   &   &   &   \\ 
J1327$-$6400 & J1327$-$6400 & 0.0031 & 5.6e+34 & 15.5 & 0.36 & --- &    &    &   &   &   &   \\ 
B2127+11F & J2129+1210F & 0.0026 & 1.9e+34 & 12.9 & --- & 0.10 & M15  &    &   &   &   &   \\ 
J1524$-$5706 & J1524$-$5706 & 0.0024 & 1.0e+34 & 11.4 & 0.41 & --- &    &    &   &   &   &   \\ 
J1216$-$6223 & J1216$-$6223 & 0.0013 & 1.3e+34 & 16.6 & 0.15 & --- &    &    &   &   &   &   \\ 
J0535$-$6935 & J0535$-$6935 & 0.00031 & 5.6e+34 & 49.4 & 0.05 & --- &  LMC &    & 3EG J0533$-$6916 & 0.59 &   &   \\ 
B0456$-$69 & J0455$-$6951 & 0.00014 & 1.2e+34 & 49.4 & --- & 0.60 &  LMC &    &   &   &   &   \\ 
\end{supertabular}
\end{scriptsize}
\end{center}
%%\caption{Gamma ray pulsar candidates (all known pulsars with $\dot E > 10^{34}$ ergs/s, except for those in the globular clusters Tuc 47 and M15 -- see text). 
%%They are ordered by $\sqrt{\dot E}/d^2$.
%Spin-down energy loss rate, radio emission properties and best distance estimates are shown.
%%Associations with known gamma ray sources and withpulsar wind nebulae (PWN) are also listed.}
\label{tab1}
%%\end{table*}
\end{landscape}
